Author's response to reviews

Title: Case report of hyperglycemic nonketotic chorea with rapid radiological resolution

Authors:

tharuka herath (tharukaherath11@gmail.com)
sithumini pahalagamage (Sithu2@yahoo.com)
sunethra senanayaka (Sunethraaes@yahoo.com)

Version: 2 Date: 05 Sep 2017

Author’s response to reviews:

1. Please consider the list of authors as it currently stands with reference to our guidelines regarding qualification for authorship (http://www.biomedcentral.com/submissions/editorial-policies#authorship).

i have done the necessary changes

2. Please improve the quality of the formatting of the figures. For example we would suggest cropping the images as much as possible so that the text is not visible and including gaps in between the panels. We would suggest that you refer to the following paper for guidance on figure formatting: https://bmcmedimaging.biomedcentral.com/articles/10.1186/s12880-017-0210-z

i improved the quality

3. On re-submission please remove the CARE checklist and cover letter files as these will no longer be required.

i removed them